Author's response to reviews

Title: Intraoperative blood loss during different stages of scoliosis surgery: A prospective study

Authors:

Hitesh N Modi (modispine@yahoo.co.in)
Seung-Woo Suh (spine@korea.ac.kr)
Jae-Young Hong (gpstylep@hanmail.net)
Sang-Heon Song (hnm7678@korea.ac.kr)
Jae-Huyk Yang (gurospine@naver.com)

Version: 4 Date: 26 June 2010

Author's response to reviews:

Apart from previously revised article, additionally we have taken help from Editing company to help us editing the manuscript as requested by reviewer and editor.
We hope this time it would fulfill the need before publication.
Thanks
Seung Woo Suh